Author’s response to reviews

Title: PTPRF is disrupted in a patient with bilateral amastia

Authors:

Surasawadee Ausavarat (ausavarat.s@gmail.com)
Siraprapa Tongkobpetch (a_siraprapa@yahoo.com)
Verayuth Praphanphoj (vpthailand@yahoo.com)
Charan Mahatumarat (charan.m@chula.ac.th)
Nond Rojvachiranonda (nond.r@chula.ac.th)
Thiti Snabboon (snabboon@yahoo.com)
Thomas C. Markello (markellot@mail.nih.gov)
William A. Gahl (bgahl@helix.nih.gov)
Kanya Suphapeetiporn (Kanya.Su@chula.ac.th)
Vorasuk Shotelersuk (vorasuk.s@chula.ac.th)

Version: 2 Date: 30 July 2010

Author’s response to reviews: see over
Dear Dr. Leigh:

Please consider the revised version of the manuscript, “PTPRF is disrupted in a patient with bilateral amastia” for publication as a “Research Article” in a future issue of BMC Medical Genetics. According to your comments, we have provided a copy of the consent form stating that the patient was aware that the medical information and pictures will be freely available on the medical journals or for other academic reasons. This work neither contains new sequences nor Gene Expression Data to be deposited. This paper provides, for the first time, evidence of PTPRF involvement in human breast development. This finding could have implications for mammalian evolution, human mammary gland development, and breast cancer.

This manuscript represents original work, is not under consideration for publication elsewhere, and has been approved by all authors.

Please address correspondence to:
Kanya Suphapeetiporn, M.D., Ph.D.
Division of Medical Genetics and Metabolism
Department of Pediatrics, Sor Kor Building, 11th floor
King Chulalongkorn Memorial Hospital
Bangkok 10330, Thailand
Tel 66-2-256-4989 Fax 66-2-256-4911
e-mail: kanya.su@chula.ac.th

For five referees, we would like to suggest as the followings:

1. Frits A. Beemer
   Department of Medical Genetics
   University Medical Center Utrecht,
   Utrecht, The Netherlands
   E-mail: F.A.Beemer@planet.nl

2. Allen E Bale
   Department of Genetics, P.O. Box 208005
   Yale University School of Medicine
   333 Cedar Street
   New Haven, CT, 06520-8005, USA
   Phone: (203) 785-5745; Fax (203) 785-7227
   E-mail: allen.bale@yale.edu
3. Maximilian Muenke, MD  
Medical Genetics Branch, National Human Genome Research Institute,  
National Institutes of Health,  
35 Convent Drive, MSC 3717, Building 35, Room 1B-203,  
Bethesda, MD 20892-3717  
USA  
E-mail: mmuenke@nhgri.nih.gov

4. Jean Claude Zenklusen, PhD  
Neuro-Oncology Branch, National Institutes of Neurological Disorder and Stroke, NIH, Bethesda, Maryland  
Bldg 37, Room 1142B  
Convent Drive, MSC4254  
Bethesda, MD 20892-4254  
Telephone: (301) 451-2144  
Fax: (301) 480-4743  
E-mail: jz44m@nih.gov

5. David I. Wilson  
Division of Human Genetics,  
Southampton University, The Duthie Building,  
Southampton General Hospital, Tremona Road,  
Southampton SO16 6YD, UK  
E-mail: d.i.wilson@soton.ac.uk

6. Wiljan J. A. J. Hendriks  
Department of Cell Biology,  
Radboud University Nijmegen Medical Centre,  
The Netherlands.  
E-mail: w.hendriks@ncmrls.ru.nl

Thank you very much for your consideration.

Sincerely,

Kanya Suphapeetiporn, M.D., Ph.D.  
Corresponding Author